Author’s response to reviews

Title: Effectiveness of Motivational Interviewing in Primary Care Patients With Dyslipidemia: A Randomized Cluster Trial. Dislip-EM Study

Authors:

JULIA BÓVEDA-FONTÁN (jboveda1@gmail.com)
NIEVES BARRAGÁN-BRUB (nieves.barragan@gmail.com)
MANUEL CAMPÍNEZ-NAVARRO (manuelcn@camfic.org)
LUIS ANGEL PERULA DE TORRES (langel.perula.sspa@juntadeandalucia.es)
JOSEP MARIA BOSCH-FONTCUBERTA (boschfontcuberta@gmail.com)
REMEDIOS MARTÍN-ALVAREZ (rmartin@perevirgili.catsalut.net)
JOSE ANTONIO PRADOS-CASTILLEJO (joseantonio.prados@gmail.com)
JUAN CARLOS ARBÓNIES-ORTIZ (juancarlos.arboniesortiz@osakidetza.net)
JESÚS MANUEL NOVO-RODRÍGUEZ (jnovor@gmail.com)
MARGARITA CRIADO-LARUMBE (margarita.criado.exts@juntadeandalucia.es)
JOSE ANGEL FERNÁNDEZ-GARCÍA (joseangelfernandezgarcia@hotmail.com)
ENRIQUE MARTÍN-RIOBOÓ (martinribooenrique@gmail.com)

Version: 2
Date: 27 January 2015

Author’s response to reviews: see over
Córdoba (Spain) 25, January, 2015

Dear sir,

We are enclosing a manuscript entitled “Validation of a Scale for Measuring Motivational Interviewing Skills in Primary Care: EVEM Study Protocol” for your consideration to be published in your Journal.

This work has not been published previously in any other peer-reviewed media (other than in abstract form) and is not currently under review elsewhere. All authors listed on the manuscript have contributed sufficiently to the project to be included as authors. Written permission has been also obtained from all persons named in the acknowledgments.

Looking forward hearing from you

Luis A Pérula de Torres, MD
Unidad Docente de Medicina de Familia y Comunitaria de Córdoba
Servicio Andaluz de Salud
C/ Isla de Lanzarote, 3
14011 Córdoba (Spain)